Author's response to reviews

Title: Limited value of long-term biochemical follow-up in patients with adrenal incidentalomas - A retrospective cohort study

Authors:

Hannah Yeomans (hannah.yeomans@gmail.com)
Jan Calissendorff (jan.calissendorff@sodersjukhuset.se)
Cristina Volpe (cristina.volpe@sodersjukhuset.se)
Henrik Falhammar (henrik.falhammar@ki.se)
Buster Mannheimer (buster.mannheimer@sodersjukhuset.se)

Version: 2
Date: 24 October 2014

Author's response to reviews: see over
Dear Mrs Cruz

We have now complied to the requested changes regarding the paper *Limited value of long-term biochemical follow-up in patients with adrenal incidentalomas*

- *A retrospective cohort study*

Hence a description of the author’s contributions, together with a consent statement is provided in the manuscript. In addition, the manuscript does now include line and page numbers.

Sincerely yours

Buster Mannheimer